Author's response to reviews

Title: Autonomous exercise game use improves metabolic control and quality of life in type 2 diabetes patients - a randomized controlled trial

Authors:

   Kerstin Kempf (kerstin.kempf@wdgz.de)
   Stephan Martin (martin@wdgz.de)

Version: 4 Date: 23 September 2013

Author's response to reviews:

There had been no further questions from the referees.